\nocite{Zhang.sep.57.347}
\nocite{Claerbout.sep.59.1}
\nocite{Zhang.sep.59.87}
\nocite{Zhang.sep.59.201}
\nocite{Nichols.sep.60.67}
\nocite{Zhang.sep.60.85}
\nocite{Zhang.sep.60.179}
\nocite{Zhang.sep.60.285}
\nocite{Zhang.sep.61.69}
\nocite{Zhang.sep.65.29}
\nocite{Zhang.sep.65.89}
\nocite{Zhang.sep.67.93}
\nocite{Zhang.sep.67.123}
\nocite{Zhang.sep.70.73}
\nocite{Zhang.sep.70.107}
\nocite{Zhang.sep.70.257}
\nocite{Zhang.sep.70.275}
\nocite{Zhang.sep.72.77}
\nocite{Zhang.sep.72.153}
\nocite{Zhang.sep.72.171}
\nocite{Zhang.sep.72.227}
\nocite{Zhang.sep.72.237}
\nocite{Zhang.sep.75.243}
\nocite{Zhang.sep.75.289}
\nocite{Zhang.sep.73.185}
\nocite{Zhang.sep.73.293}
\nocite{Zhang.Phd.93}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  26 
articles.
{\footnotesize
\putbib[SEP]
}
